Author's response to reviews

Title: Specific genes involved in synthesis and editing of heparan sulfate proteoglycans show altered expression patterns in breast cancer

Authors:

Iván Fernández-Vega (ivan_fernandez_vega@hotmail.com)
Olivia García (garciaolivia@uniovi.es)
Ainara Crespo (acrespo@neiker.net)
Sonia Castañón (scedelatorre@neiker.net)
Primitiva Menéndez (tivamenendez@telecable.es)
Aurora Astudillo (astudillo@hca.es)
Luis M Quirós (quirosluis@uniovi.es)

Version: 5 Date: 10 December 2012

Author's response to reviews:

Editorial Office
BMC CANCER

Dear Sirs:

Please find enclosed a revised version of our manuscript “Specific genes involved in synthesis and editing of heparan sulfate proteoglycans show altered expression patterns in breast cancer “. The text has been reviewed by a native English speaking colleague and the corrections made are highlighted in red.

Yours sincerely,

Dr. Luis M. Quirós
Departamento de Biología Funcional. Facultad de Medicina.
C/Julián Clavería s/n, 33006-Oviedo, SPAIN.
Tel.: 985103000 ext.: 5213
E-mail: quirosluis@uniovi.es